Author’s response to reviews

Title: The utility of transbronchial rebiopsy for peripheral pulmonary lesions in patients with advanced non-squamous non-small cell lung cancer

Authors:

Akiko Tateishi (akikotoku1002@gmail.com)
Yuji Matsumoto (yumatsum@ncc.go.jp)
Midori Tanaka (midotana@ncc.go.jp)
Toshiyuki Nakai (m2065051@med.osaka-cu.ac.jp)
Shinji Sasada (sasastaf@hotmail.co.jp)
Masahiro Aoshima (aolung1@nifty.com)
Takaaki Tsuchida (ttsuchid@ncc.go.jp)

Version: 3 Date: 26 Aug 2020

Author’s response to reviews:

We thank the editor and reviewers for reviewing and critiquing our manuscript, as their valuable comments have helped to improve our manuscript. We hope that our revised manuscript is acceptable for publication in your journal. The changes and corrections made to the original manuscript are marked in yellow highlight. In addition, English was reviewed and proofread by an English native speaker. Please find the editing certificate attached to the supplementary material.

Comments from Editor:

English should be reviewed by an English native speaker.
→English was reviewed and proofread by an English native speaker. Please find the editing certificate attached to the supplementary material.

Abstract

Abbreviations in abstract should be avoided if possible or clarified since not all readers will understand what R-EBUS means. In the abstract conclusions, I would say feasible rather than practical.
→Thank you for your comment. We have rewritten the sentence as below.
Page 2, lines 5-6: “...using bronchoscopy with radial endobronchial ultrasound guidance in patients with advanced NSCLC remain unclear.”
Page 2, line 23: practical →feasible
Once again, we would like to thank Dr. Ernest Nadal (Editor-in-Chief) for your review of our manuscript and the consideration of our manuscript for publication in BMC Pulmonary Medicine.

Sincerely,

Akiko Tateishi, MD